Dear Editor:

We appreciate the thoughtful reviews of our manuscript and the insightful comments. We have responded to each of the comments of these editors as much as possible. The corrections in the paper and the responses to the editor’s comments are as following:

Company Editor-in-Chief

Jin-Lei Wang

I have reviewed the Peer-Review Report, the full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. Before its final acceptance, the author(s) must provide the Signed Informed Consent Form(s) or Document(s) of treatment. For example, authors from China should upload the Chinese version of the document, authors from Italy should upload the Italian version of the document, authors from Germany should upload the Deutsch version of the document, and authors from the United States and the United Kingdom should upload the English version of the document, etc. The author(s) must provide the English Language Certificate issued by a professional English language editing company. Please visit the following website for the professional English language editing companies we recommend: https://www.wjgnet.com/bpg/gerinfo/240.
Response: We appreciate the insightful comments and the suggestions by Editor Wang. We have uploaded the Signed Informed Consent Forms of treatment in Chinese. We used the English editing services from the “American Journal Experts (AJE)” as you recommended. The editorial certificates were shown below.

Science Editor

Man Liu

1 Scientific quality: The manuscript describes a Case Report of the gastrointestinal stromal tumors. The topic is within the scope of the WJCC. (1) Classification: Grade B; (2) Summary of the Peer-Review Report: Good and the research provides sample data for the authors to make their conclusion. Grammar need Some revision. The questions raised by the reviewers should be answered; (3) Format: There are 3 figures; (4) References: A total of 21 references are cited, including 3 references published in the last 3 years; (5) Self-cited references: There is no self-cited reference. 2 Language
Evaluation: Classification: Grade C. Academic norms and rules: The authors provided the Written informed consent. No academic misconduct was found in the Bing search. There is a preprint version of the manuscript. Supplementary comments: This is an unsolicited manuscript. The study was supported by Healthy department of Hainan province and Hainan Provincial Nature Science Foundation of China. The topic has not previously been published in the WJCC. Issues raised:

1) The language classification is Grade C. Please visit the following website for the professional English language editing companies we recommend:

https://www.wjgnet.com/bpg/gerinfo/240

Response: We appreciate this suggestion. We used the English editing services from the “American Journal Experts (AJE)” as you recommended. The editorial certificates were shown below.

[Image of AJE Editing Certificate]

This document certifies that the manuscript

*Paraneoplastic focal segmental glomerulosclerosis associated with gastrointestinal stromal tumors with cutaneous metastasis: A case report*

was edited for proper English language, grammar, punctuation, spelling, and overall style by one or more of the highly qualified native English speaking editors at AJE.

This certificate was issued on July 18, 2021 and may be verified on the AJE website using the verification code 071E-9282-86F8-859C-2861.

Neither the research content nor the authors' intentions were altered in any way during the editing process. Documents receiving this certification should be English-ready for publication; however, the author has the ability to accept or reject our suggestions and changes. To verify the final AJE edited version, please visit our verification page at ajel.com/certificate. If you have any questions or concerns about this edited document, please contact AJE at support@aje.com.

AJE provides a range of editing, translation, and manuscript services for researchers and publishers around the world. For more information about our company, services, and partner discounts, please visit ajel.com.
(2) The authors did not provide the approved grant application form(s). Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

Response: We appreciate this suggestion. We have uploaded the approved grant application forms.

(3) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor;

Response: We appreciate this suggestion. We have provided the original figure documents using PowerPoint.

(4) PMID and DOI numbers are missing in the reference list. Please provide the PubMed numbers and DOI citation numbers to the reference list and list all authors of the references. Please revise throughout;

Response: We appreciate this suggestion. We have provided PMID and DOI numbers in the reference list and listed all authors of the references.

(5) The “Case Presentation” section was not written according to the Guidelines for Manuscript Preparation. Please re-write the “Case Presentation” section, and add the “FINAL DIAGNOSIS”, “TREATMENT”, and “OUTCOME AND FOLLOW-UP” sections to the main text, according to the Guidelines and Requirements for
Manuscript Revision;

Response: We appreciate this suggestion. We have re-write the “Case Presentation” section, and add the “FINAL DIAGNOSIS”, “TREATMENT”, and “OUTCOME AND FOLLOW-UP” sections to the main text, according to the Guidelines and Requirements.

(6) The authors need to download and fill out the CARE Checklist (2016) with page numbers

Response: We appreciate this suggestion. We have filled out the CARE Checklist (2016) with page numbers.

(7) Please upload the primary version (PDF) of the Informed Consent Form (Surgical procedures or other) that has been signed by the patients in the study, prepared in the official language of the authors’ country to the system; for example, authors from China should upload the Chinese version of the document, authors from Italy should upload the Italian version of the document, authors from Germany should upload the Deutsch version of the document, and authors from the United States and the United Kingdom should upload the English version of the document, etc. Example: Download our sample of signed informed consent-Case report, at https://www.wjgnet.com/bpg/GerInfo/287. Note: To obey the publication ethics and improve the protection of all patients’ rights to privacy, the authors should provide the informed consent form on which the patient's name, address, birthday, address, ward,
bed number, hospital number and other private information are obfuscated.

Response: We appreciate this suggestion and have uploaded the Signed Informed Consent Forms of treatment in Chinese.

6 Recommendation: Conditional acceptance.